Author’s response to reviews

Title: Positive melioidosis serology in a patient with adult onset Still's disease: A case report of a diagnostic dilemma

Authors:

Harsha Dissanayake (dissanayakeha@gmail.com)
Gayani Premawansa (gavisprema@gmail.com)
Enoka Corea (enokacorea@hotmail.com)
Inoshi Atukorala (inoshi.atu@gmail.com)

Version: 1 Date: 07 Sep 2018

Author’s response to reviews:

A detailed response letter is attached